Author's response to reviews

Title: Seroprevalence of anti-hepatitis E virus (HEV) in a Korean population: comparison of two commercial anti-HEV assays

Authors:

Hyun Kyung Park (totoro1128@gmail.com)
Sook-Hyang Jeong (jsh@snubh.org)
Jin-Wook Kim (kimiw@snubh.org)
Byung-Hyun Woo (id791128@naver.com)
Dong Ho Lee (dhljohn@yahoo.co.kr)
Hyun Young Kim (cecilup@naver.com)
Soyeon Ahn (35011@snubh.org)

Version: 4 Date: 29 May 2012

Author's response to reviews: see over
26\textsuperscript{th} May, 2012

Dear editors of \textbf{BMC Infectious Diseases}

MS:1017726961679798

Title: \textit{Seroprevalence of anti-hepatitis E virus (HEV) in a Korean population: comparison of two commercial anti-HEV assays}

We are grateful the editors and reviewers of the ‘\textit{BMC infectious disease}’ for taking time and efforts to review our paper. We hope the revised manuscript will better meet the requirements of the ‘\textit{BMC infectious disease}’ for publication.

Here below is one-by-one response to your comments.

The Response to the editorial requirements:

After reading through your manuscript, we feel that the quality of written English needs to be improved before the manuscript can be considered further.

- This manuscript was reviewed to ensure quality and accurate translation by an expert who is a native speaker of English in accordance with your advice. And we added a file of proofreading certificate as below.
We thank again to evaluation of the editors and reviewers, which have allowed us to improve and clarify the manuscript.

Sincerely yours,

Sook-Hyang Jeong, M.D.

Associate Professor

Department of Internal Medicine, Division of Gastroenterology

Seoul National University Bundang Hospital
Seongnam, Republic of Korea

Tel: 82-31-787-7034  Fax: 82-31-787-4052

E-mail: jsh@snubh.org